\documentclass[12pt]{article}

\usepackage{mathrsfs}
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{graphicx}
\usepackage{mathrsfs}
\usepackage{fancyhdr}
\usepackage{lastpage}
\usepackage{ifthen}
\usepackage{amscd}
\usepackage{cancel}
\usepackage{amssymb}
%\usepackage{polynom}
\usepackage{wasysym}
\usepackage{bbm}
\usepackage{rotating}
\parindent 0 in

\topmargin-0.3in \textheight9.5in \oddsidemargin-0.3in
\textwidth7.3in

\usepackage{tikz}
\usetikzlibrary{positioning, scopes} 			%always use these
\usetikzlibrary{backgrounds}

\pagestyle{fancyplain} \lhead{Homework 0}
\chead{Math 3345, Spring 2022, Kutler}
\rhead{Due: 1/14/22}


\begin{document}

Please tell me a little bit about your self. (Apart from your name, all information is voluntary.)
\vspace{.3in}

Legal name (as it appears in Canvas):

\vspace{12pt}

Preferred name:

\vspace{12pt}

Preferred pronouns (e.g., ``she''):

\vspace{12pt}

Year in school:

\vspace{12pt}

Academic major:

\vspace{12pt}

Hometown:

\vspace{12pt}

Other math courses you've taken:

\vspace{12pt}

Why are you taking Math 3345?

\vfill

Hobbies:

\vfill

Do you have any conflicts with my office hours? Does another time work better for you?

\vfill

Is there anything else you'd like me to know?

\vfill

Do you have any questions for me?

\vfill


\end{document}
